Author's response to reviews

Title: Occurrence and quality of anticoagulant treatment of chronic atrial fibrillation in primary health care in Sweden: a retrospective study on electronic patient records

Authors:

Dr Gunnar H Nilsson (gunnar.nilsson@nlpo.sll.se)
Ingela Bjorholt (ingela.bjorholt@telia.com)

Version: 3 Date: 6 Feb 2004

Dear Editor,

Attached is a revised, and hopefully final, version of the manuscript entitled "Occurrence and quality of anticoagulant treatment of chronic atrial fibrillation in primary health care in Sweden: a retrospective study on electronic patient records ", which we would be pleased if you accept for publication in BMC Clinical Pharmacology.

As far as we can see, we have successfully addressed the two comments from the reviewer Alan Go, as itemised below.

Further, we have addressed the formatting changes requested, concerning the Tables.

Yours sincerely,

Gunnar Nilsson, MD, PhD
Family Medicine Stockholm
Karolinska Institutet
Sweden

Revisions and explanations as related to the comments in the reviewer's report dated 31 January 2004.

Major compulsory revision
The assertion that there was a gender difference in use of warfarin favouring men is now removed from the Conclusion section of the Abstract and Discussion Section.

Minor compulsory revision
In the Discussion Section, third paragraph, second sentence: The conclusion concerning the underuse of warfarin is now changed to "...our figures suggest a minor underuse."